Author's response to reviews

**Title:** Health-related quality of life after myocardial infarction is associated with level of left ventricular ejection fraction

**Authors:**

Kjell I Pettersen (kjell.pettersen@medisin.uio.no)
Elena Kvan (elena.kvan@rikshospitalet.no)
Arnfinn Rollag (arnfinn.rollag@ahus.no)
Knut Stavem (knut.stavem@medisin.uio.no)
Aasmund Reikvam (asmund.reikvam@medisin.uio.no)

**Version:** 4 **Date:** 27 September 2008

**Author's response to reviews:** see over
Dear Assistant editor

Enclosed you will find the accepted version of the manuscript:

*Health-related quality of life after myocardial infarction is associated with level of left ventricular ejection fraction*

Manuscript ID 1792412020186450

We would like to thank the three reviewers for their useful comments which improved our manuscript and the editor for accepting the revised manuscript for publication. We are also thankful for help from the BMC with fitting the tables into the right format.

We have read the accepted manuscript carefully and found some minor errors which we have corrected. All changes are listed below. In the manuscript deleted text is red with strikethrough. Added text is red and underlined.

Page 1:
A colon is deleted in the second line.

Page 2:
A comma is added in paragraph two, line three.

Page 6:
The letter “e” is added to “stat” in the last line.

Page 7:
The word “predictors” is deleted in line one and “multicolinearity” is deleted in line 3 of the third paragraph. “$R^2$” is deleted both in line one and two of paragraph four.

Page 9:
The words “of” and “score” is added in the last sentence.

Page 17:
Names of authors have been corrected in reference 1 and 2.
Page 19:
The name of one of the authors and the title have been corrected in reference 16

Page 25
The table legend text has been moved to below Table 3. We have changed the format of “R-square” in the last line in the first column of Table 3 to fit the format of the rest of the table.

All authors have read and accepted the final version of the revised manuscript.

Corresponding author: Kjell I. Pettersen,

On behalf of the authors,
Yours sincerely

Kjell I. Pettersen, MD